Author’s response to reviews

Title: Feasibility and antihypertensive effect of replacing regular salt with mineral salt- rich in magnesium and potassium- in subjects with mildly elevated blood pressure

Authors:

Essi S Sarkkinen (essi.sarkkinen@foodfiles.com)
Mika J Kastarinen (mika.kastarinen@fimea.fi)
Tarja H Niskanen (tarja.niskanen@foodfiles.com)
Pia H Karjalainen (pia.karjalainen@foodfiles.com)
Taisa M Venäläinen (taisa.venalainen@uef.fi)
Jay K Udani (jay.udani@medicusresearch.com)
Leo K Niskanen (leo.niskanen@ksshp.fi)

Version: 5  Date: 29 August 2011

Author’s response to reviews: see over
August 29th, 2011

The Nutrition Journal Editorial Team

Ref: MS: 7900875585245861 - Feasibility and antihypertensive effect of replacing regular salt with mineral salt- rich in magnesium and potassium- in subjects with mildly elevated blood pressure/email 24 August 2011

To whom it concerns,

We refer to your request dated 24 August 2011 and have formatted the manuscript according to the journal style and your instructions.

All requested major revisions are made and listed below:

The trial registration number is mentioned in the abstract. The last section of the abstract is titled ‘Trial Registration’ and list the trial registry and the unique identifying number in the following format: ISRCTN12345678. We also updated the abstract in the submission system to ensure it remains identical to the manuscript. Trial registration number is omitted from the title page.

The reference list is numbered for ease of citation. References are cited in the text using consecutive numbers in square brackets and the references are placed before the full stop.

We have made also the requested minor revisions like:

Line numbers removed.

Running title removed.

Statement of funding transferred to the ‘Acknowledgements’ section and removed from the title page.

All highlighting are removed from the manuscript.

Titles and numbering are removed from the image files. The legend and title should be part of the manuscript file after the reference list.

I hope loaded proof is acceptable for publication in the Nutrition Journal.

We hope to hear from you in the near future.

Your sincerely

Essi Sarkkinen, Ph.D.
Managing and research director
Oy Foodfiles Ltd
Neulaniemientie 2 L 6
FI- 70210 Kuopio, Finland
telephone:+358-(0)44- 7001 201
telefax: +358- (0)17- 288 1269
essi.sarkkinen@foodfiles.com